Author’s response to reviews

Title: Pancreatic hamartoma: A case report and literature review.

Authors:
DAISUKE MATSUSHITA (m7762@m2.kufm.kagoshima-u.ac.jp)
Hiroshi Kurahara (hiroshik45@gmail.com)
Yuko Mataki (mataki@m.kufm.kagoshima-u.ac.jp)
Kosei Maemura (kmaemura@m3.kufm.kagoshima-u.ac.jp)
Michiyo Higashi (east@gmail.com)
Satoshi Iino (iino@m2.kufm.kagoshima-u.ac.jp)
Masahiko Sakoda (sakoda@m.kufm.kagoshima-u.ac.jp)
Hiroyuki Shinchi (shinchi@m.kufm.kagoshima-u.ac.jp)
Shinichi Ueno (ueno1@m.kufm.kagoshima-u.ac.jp)
Shoji Natsugoe (natsugoe@m2.kufm.kagoshima-u.ac.jp)

Version: 2 Date: 14 Dec 2015

Author’s response to reviews:
Dear editor,

We really appreciate to give us to revise this manuscript again. We have made revised points in the red words as you mentioned.

And remove the consent form from the additional files.

We hope this revised manuscript will be improved for the publication of your journal.